# THE EDUCATION OF KOLHI COMMUNITY IN SINDH: OPPORTUNITIES AND CHALLENGES



## **AAKASH**

Department of Sociology

Quaid-i-Azam University Islamabad

# THE EDUCATION OF KOLHI COMMUNITY IN SINDH: OPPORTUNITIES AND CHALLENGES



"Thesis submitted to the Department of Sociology, Quaid-i-Azam
University, Islamabad, for the partial fulfillment of the degree of
Master of Science in Sociology"

By

Aakash

Department of Sociology

Quaid-i-Azam University Islamabad

#### Acknowledgment

Firstly, I am thankful to my family who supported me financially and morally to complete my thesis. Secondly, I am thankful to my supervisor Dr. Hazir Ullah, who guided me whenever I felt difficulties. Honestly, he is the person from whom I have learned a lot about research work and how to address the issues that exist in our society by doing proper research on it.

Moreover, I am thankful to all the people who helped me in my thesis, fieldwork, and interview guide, here are some of them, Teerath Kolhi, Ajmal Kolhi, and Teko Mal Kolhi, who helped me to find respondents of the community, and took me to the different villages near Badin and Malkani city, where most of the population of Kolhi community resides.

Aakash

#### **Abstract**

The majority of the people from scheduled castes in Pakistan are uneducated and are engaged in agriculture. They work for the feudals. This study attempts to know about the educational experiences and challenges that student from the scheduled caste Hindu Kolhi communities face at the school level in Pakistan. The main purpose of the current study is to identify the reasons for less education in a Dalit community as well as the major challenges that they face while going to school. The current study was conducted in various villages of the district Badin, Sindh, through face-to-face interactions with students in grades 6 to 10. A quantitative approach was used to address the issues with marginalized community 'Kolhi' regarding their education. Moreover, random sampling techniques were used for the selection of sample size, and data were analyzed using Statistical Package for Social Sciences (SPSS). The findings of this study reveal that majority of the Kolhi community does not have schools close to their homes due to their residence and occupation in rural areas. Most of the students argued that they are being differentiated by teachers and other students by using separate pots for eating and glasses for drinking at school, which creates a lack of interest among the students who go to school. The study also found that early marriages of children between the ages of 10 and 14 which is an old tradition of the community, also hinders them to continue further education.

## **Table of Contents**

Chapter No: 1	1
INTRODUCTION	1
1.1 Statement of Problem	8
1.2 Objectives	8
1.3 Research Questions	8
1.4 Hypothesis	9
1.4.1 Null Hypothesis (H <sub>o</sub> )	9
1.4.2 Alternative Hypothesis (H <sub>1</sub> )	9
1.5 Significance of the Study	9
1.6 Organization of the thesis	10
Chapter No: 2	11
LITERATURE REVIEW	11
Chapter No: 3	33
RESEARCH METHODOLOGY	33
3.1 Research Design	34
3.2 Locale of the study	35
3.3 Target Population	35
3.4 Sampling techniques	35
3.5 Sample Size	35
3.6 Tools for Data Collection	35
3.7 Techniques for Data Collection	36
3.8 Tools for Data Analysis	36
3.9 Ethical Concern	36
Chapter No: 4	37
FINDING AND RESULTS	37

Chapter No: 5	58
DISCUSSION, CONCLUSION, AND RECOMMENDATION.	58
5.1 Discussion	59
5.2 Major Findings	66
5.3 Conclusion	68
5.4 Recommendations	69
5.5 Limitations of the study	70
REFERENCES	71
ANNEXURE	85

## **List of Tables**

Table 4.1. Age of the respondents
Table 4.2. Education level of the respondent
Table 4.3. Occupation or Profession of the respondent40
Table 4.4. The school is quite away from your home41
Table 4.5. Your parents want you to get a proper education42
Table 4.6. Your family has enough money to support your education43
Table 4.7. You work as labor after school time44
Table 4.8. Your family has enough money to buy books and copies 45
Table 4.9. Your parents have financial support to send you for secondary
education46
Table 4.10. Your parents want that you should work with them on the land.
47
Table 4.11. Your parents' expectation after not getting any job48
Table 4.12. Due to a lack of job opportunities, you prefer to work on
agricultural land
Table 4.13. Your family migrates from one place to another for seasonal
earning50
Table 4.14. You cannot attend the school continuously, due to seasonal
migration
Table 4.15. Do you think that your parents do your marriage during your
studies at the school level?
Table 4.16. Your teachers discriminate against you based on your religion.
53
Table 4.17. Your teachers do not respond to you then they respond to
Muslim fellows. 54
Table 4.18. You feel discrimination from your Muslim fellows in class55
Table 4.19. Your glass and plates are separate at the school canteen56
Table 4.20. You are allowed to eat at the same food points in school57

Chapter No: 1

INTRODUCTION

Education is the main resource to the nation's development of any country in terms of political, social, and economic conditions. It makes a feeling of self-responsibility among individuals and communities. "Through education, people not only realize their duties but also they can understand how to achieve their national, societal and individual rights" (Ahmed et all. 2014). Educational issues and problems are mainly found in the developing countries in the world including Pakistan. However, the issues with marginalized Hindu communities, particularly the lower castes (scheduled castes) Kolhi community, are the most severe in lower southern Sindh.

Kolhis primarily belong to the Hindus ethnic group, though, some are Christian and Muslim who were changed over in 1940-50 and occupied with farming land. The majority of them are sharecroppers, farmers, labor bounded and landless ethnic groups.

"Most of the untouchables (Dalit) communities are bounded in structured ethnic discrimination and socio-religious exploitation", Kolhi community is one of them (Hussain 2014). The main problem behind labor bounded and landless peasant is the quality of education they get. They are more victims of deprived and excluded from basic resources like education. They mostly work as farmers for Muslim landlords and live as landless peasants. Kolhi community is a marginalized community, having less literacy rate, high poverty rate, and less socially developed community. Therefore, the present study aimed to identify the reasons behind the lack

of access to education of Dalit children especially Kolhi's in the schools, and the major challenges they face to get the proper education. There are a number of other communities that come under scheduled caste such as Bheel, Meghwar, Oad, Walimki, and Bhangi, however, in this study particularly the Kolhi community was chosen for the research population living in district Badin, Sindh.

A study conducted by Karade (2008) revealed that "the backwardness of Dalit communities will be shown from their illiteracy, Joblessness and dependent on agriculture and social segregation". In this modern world, education plays a crucial role to boost up the best way of living life and improving the socio-economic position in society, and there is a lack of both things in the Kolhi community. As Chowdhury (2009) defines that historically Dalits are found victims of social exclusion and discrimination that limit them to access education. Therefore, the current research has tried to answer the question; what are the major reasons behind labor bounded and landless peasants of the Kolhi community getting proper education? The Dalits also called scheduled castes or untouchables "have experienced predictable denial to access to education since the 1850s (Fraser 2010). As Kolhi that are living in Sindh have not made much progress and changes towards the modern way of living life and to have the basic rights as like education. In this time "having a good education helps people to think and to live a better life in the society; it improves not only an individual but also their whole community through new ideas and social skills" (Al-Shuaibi 2014). By developing human capital, education

adds to a country's economic growth (Abuya et al. 2014). Higher-education countries are more likely to be politically stable, have better chances of survival, and achieve significant health gains (USAID 2008). It bridges the gap between religious minorities and opportunities, which contributes to their weak socio-economic status. On the other hand, the study cited above, while failing to provide a comprehensive picture of the various groups living in the slums to a large extent, covers the inequalities within the slums, leading to their contradictions for specific groups (Samuel and Nisar 2020). According to Shah (2007) due to a lack of equity, property ownership, and economic ventures, the scheduled castes rely on wage labour for their livelihood. The bulk of them live and work as agricultural workers in rural areas, with some working as daily wagers in nearby towns and cities. They are frequently forced to perform menial activities that are not performed by Muslims or upper-caste Hindus.

Kolhi peasant activism provides a unique opportunity to understand peasant activism from the perspective of their education and labor bounded of lower caste, the ethnically discriminated and marginalized minority community. They are debarred from the greater part of the conventional graces of life (Béteille 2003 as Cited in Deen 2015). Existing literature on Dalit education shows that very few Dalits tend to continue to a certain level of education that could create value for their livelihood. As my research objective is to find out the reasons behind Dalits children's lack of access to education, and the challenges or opportunities they face in Educational Institutions, thus I choose the Kolhi Community in Sindh,

who is the more victim of having lack of access to education. Education is an important means of social mobility (Bates 1998; Mabeya 2018). The limited studies on religious minorities in slums that have been published demonstrate a complicated interplay of social, economic, cultural, and physical difficulties that prevent religious minority children from attending school in developing countries (Hossain 2016; Husain 2005)

In this study, the term 'Scheduled caste' and 'Dalit' refers to those who are socially excluded discriminated against, on the ground of caste and (stigmatized) occupations. The focus of the study is the Non-Muslim Hindu Dalit community, who are generally known as a landless ethnic group or marginalized community (Kolhi) living in the backward areas of Sindh, preferably in the *Laar* side (lower Sindh). Most of the youth in the Kolhi community, work as a labor in the land with the whole family from their childhood "which creates adulthood unemployment and lack of education in youth" (Naz et al. 2014). Furthermore, according to International Labor Organization (ILO) (2005 & 2006) as cited by Naz et al. (2014), child labor is a work that is associated to mentally, physically, socially, and morally harmful to children and affects their schooling by depriving them of the opportunity to have education, obliging children to leave school prematurely or requiring them to combine school attendance with an excessively long and heavy workload.

According to a survey, illiteracy affects two-thirds of the scheduled caste people in southern Punjab and Sindh. In Umerkot, Tharparkar, Rahimyar

Khan, and Bahawalpur, statistics suggest that 74 percent of scheduled castes are illiterate. Only 4% of the 26 percent who claimed to be literate have completed secondary school, compared to 15% who have completed primary school, 4% who have completed middle school, and another 4% who have completed Matric. Only 1% of the scheduled castes have graduated, and only a few are fortunate enough to have post-graduate degrees. The survey also supports scheduled caste communities' and leaders' claims that they are at the bottom of all social development indices and that government development initiatives are biassed against them (Shah 2007). Furthermore, Scheduled caste students and parents have alleged that their peers and teachers make fun of them because of their poor clothing and appearance. Many of the students in this group wore ragged clothes and were barefoot. Unfortunately, rather than integrating and promoting excluded groups into mainstream society, the entire system appears to alienate them even further, and there appears to be no systematic effort to reduce prejudice and discrimination.

Meghwar community is also a scheduled caste but has developed the living style and educational level toward modern life, as compared to Bheel and Kolhi communities. Although the Meghwar community also felt discrimination from upper-caste Hindus but struggled against them, and now the community is going shoulder by shoulder with upper caste people through education. Research in such a social environment can help in the development of new ideas. These cultural inequalities often do not provide

opportunities for testing and expanding the theories that have emerged from research on industrialized countries (Buchmann and Hannum 2001).

A study by Hossain (2016) on Dalits in Bangladesh found that all of the Dalit children, who were interviewed, had been humiliated by their senior students, either directly or indirectly. Dalit children are usually placed in groups of 32 at educational facilities. They are treated with scorn and humiliation by students from mainstream culture, making them feel uncomfortable and powerless, and they develop a sense of inferiority. Because of their uneasiness and inferiority complex, they typically enroll in school as a group. When his or her colony's buddies drop out, a solitary child rarely takes a class and left school. Due to caste-based shame and a fear of physical insecurity, they are unable to attend school alone. As a result, when people are alone, they tend to give up. The ability of Dalit parents to invest in their children's education is constrained by their income poverty as a result of being restricted to menial jobs. Low parental education and employment status have a negative impact on children's education since both factors affect income and awareness levels. Parents' inadequate educational skills can increase the cost of their children's education because youngsters receive minimal help from their parents with schoolwork, necessitating expensive private tuition. Unsuitable family conditions make it difficult or impossible for them to study at home, and cultural factors worsen the situation. Girl children's education is impacted by a lack of basic amenities, child marriage, insecurity, fear of defamation, gender bias, and traditional ideas about their destiny (Hossain 2016).

#### 1.1 Statement of Problem

Education is the basic right of every individual, but most people are deprived of it and bound by ethnic discrimination, which leads to a high poverty rate and less literacy rate. Education is a challenge for Dalit communities who are marginalized and work as sharecroppers. Kolhi Community is one of them (Scheduled castes) that "works as sharecroppers and labor bounded under Muslim landlords and feudalists" (Hussain 2014). Mostly they are less educated and poor living style as landless peasants in Sindh. The main reason behind labor bounded and landless peasant is lack of access to education. Kolhi Community is facing these challenges nowadays. For their betterment and progression in living a proper life, without bounded in ethnic discrimination, we need to study more about them, that what are the reasons behind lack of access to education? And how to engage them in educational institutions and various awareness programs to build up fundamental rights.

#### 1.2 Objectives

- To find out the challenges faced by the Kolhi community in getting education.
- To find out the main reasons behind the lack of access to education in the Kolhi community.

#### 1.3 Research Questions

 Why do scheduled caste communities have little access to education in Sindh?  What are the challenges faced by scheduled castes children in attending school?

### 1.4 Hypothesis

This study analyzes the structural problems and discrimination in education, with the marginalized Kolhi community in Sindh.

### 1.4.1 Null Hypothesis (H<sub>o</sub>)

There is no relationship between structured problems, discrimination, and education of the Kolhi community.

#### 1.4.2 Alternative Hypothesis (H<sub>1</sub>)

There is a relationship between structured problems, discrimination, and the education of the Kolhi community.

#### 1.5 Significance of the Study

This study would be beneficial for the government to address the educational problems and issues with marginalized communities who are ethnically classified in every sphere of life. This research has highlighted the issues and reasons for high illiteracy among the Kolhi community. It has spotlighted the problems faced by youth to get a proper education.

There is very little literature available on community-based educational issues and problems. This study highlights the hidden problems and issues with marginalized and socially excluded communities. This research can be beneficial for future researchers to work on the educational challenges

of poor communities, and to study how they are socially deprived of basic rights like Education.

Existing literature on the Kolhi community shows the poor socioeconomic condition of the community but no one has talked that what are the main reasons behind the poor conditions. Therefore, this study shows a relationship between poor economic conditions and education. Due to the high illiteracy rate and unawareness of basic rights they are ethnically bounded and victimized under the feudalist and landlords.

#### 1.6 Organization of the thesis

Chapter one gives the introduction to the topic and explains the objectives of the study, statement of problems, research questions, the significance of the study, and thesis outline.

Chapter two consists of the literature review related to the topic.

Chapter three describes the research methodology used to conduct the study including tools and techniques for data collection, the locale of the study, analysis of data, and ethical consideration is discussed.

Chapter four reports the findings in the form of tables and a description of tables in detail.

Chapter five discusses the major findings and summarizes the conclusion based on the findings of the study. Moreover, the limitation of the study, major key findings, and recommendations are also presented for future researchers in this field.

Chapter No: 2

## LITERATURE REVIEW

Pakistan has one of the world's lowest literacy rates. Pakistan's literacy rate is 52 percent, with just 80 million individuals out of 140 million, able to read and write, according to official data. The majority of the illiterate population in these places is made up of women, children, religious minorities, low caste, bonded labourers, and other vulnerable and marginalized groups (Shah 2007). Education is the main source for every individual or a community to change their socio-economic conditions towards the modern world and to reduce the poverty rate and unemployment rate by getting a good education. It makes a feeling of selfresponsibility among individuals. Through education, people not only realize their duties but also can understand how to achieve their national, societal, and individual rights (Ahmed et all. 2014). And "very little literature is found on the inequalities with lower castes communities in education" (Samuel and Nisar 2020) in Pakistan. Kolhi community is one from the list of lower castes (scheduled castes) who are less educated and less socio-economically developed, as compared to Meghwar, Bagri, Valimiki, and Oad in Sindh (Hussain 2014). Scheduled caste students and parents have alleged that their peers and teachers make fun of them because of their poor clothing and appearance. Many of the students in this group wore ragged clothes and were barefoot. Unfortunately, rather than integrating and promoting excluded groups into mainstream society, the entire system appears to alienate them even further, and there appears to be no systematic effort to reduce prejudice and discrimination (Shah 2007). Pakistan's National Council for Justice and Peace (NCJP) reports that the

literacy rate among scheduled caste Hindus is lower (19 percent) as compared to upper-caste Hindus (34 percent), compared to the national average of 46.56 percent. And if we divide the 19% within scheduled castes, it will be less than the 1% literacy rate of the Kolhi community in Sindh. "There are around 40 castes in Pakistan, 32 of them were recorded as scheduled castes under the November 1957 Presidential law of Pakistan" (Patel 2020). Kolhi is one of these recorded scheduled castes, alongside Bheels, Meghwars, Baghris, and others. A survey conducted in 2014 in Bangladesh by Islam et al. (2015) reveals that "Dalit communities are extremely poor and significantly vary in literacy rate". Moreover, Ullah and Ali (2018) pointed out that educational expectations of children determined by their primary education influence and socio-economic conditions of the family on the reproduction of class hierarchies. The educational backwardness of the Dalit communities is attributed to Poverty and illiteracy (Nambissan 1996). Kolhi community is land-less peasants who are working in agriculture and largely as sharecroppers under landlords and feudalists for their daily wages.

Kolhi community is generally a Hindu, however, some of them are Christian and Muslim who were changed over in 1940-50 and occupied with farming land, and mostly they are sharecropper, farmers, labor bounded and landless ethnic groups. Kolhi community is also called Koli, they are mostly residing in the area of Lower Sindh (Laar), and some are in the Nagarparkar Tharparkar. They engaged in Agriculture and are largely under labor bounded, feudalism, poor, and sharecroppers (Hussain

2014). As per the Global Slavery Index, practically 45.8 million individuals are caught in labor bounded across the world and Pakistan is home to 2 million of them. Bonded labor was prohibited in Pakistan in 1992, however, it is yet found in brick kilns, and also in the agriculture, mining, and fishing areas (Ghani 2017). Low-castes were subjected to various forms of spatial exclusion that 'passively' (Sen 2000) prevented them from gaining access to educational opportunities (Tamim and Tariq 2015). Furthermore, according to International Labor Organization (ILO) (2005 & 2006) as cited by Naz et al. (2014), child labor is a work that is associated to mentally, physically, socially, and morally harmful to children and affects their schooling by depriving them of the opportunity to have education, obliging children to leave school prematurely or requiring them to combine school attendance with an excessively long and heavy workload. These people are traditionally engaged with various modest jobs and "historically separated in each circle of life - social, economic, political even religiously" (Hossain 2016), therefore they are most discriminated people, and "from the 1870s, these people begin to term as 'Depressed Classes'" (Charsley and Karanth 1998: 21 as cited in Hossain 2016). As per the discoveries of Thardeep Rural Development Program (TRDP), an NGO working in Thar, a greater part of the freed bonded laborers living in camps close to Hyderabad are scheduled caste Hindus. There are an expected 1.7 million reinforced workers in Sindh and a staggering larger part of them is from scheduled caste (as cited in Shah 2007). It bridges the gap between religious minorities and opportunities,

which contributes to their weak socio-economic status. On the other hand, the study cited above, while failing to provide a comprehensive picture of the various groups living in the slums to a large extent, covers the inequalities within the slums, leading to their contradictions for specific groups (Samuel and Nisar 2020).

According to a survey, illiteracy affects two-thirds of the scheduled caste people in southern Punjab and Sindh. In Umerkot, Tharparkar, Rahimyar Khan, and Bahawalpur, statistics suggest that 74 percent of scheduled castes are illiterate. Only 4% of the 26 percent who claimed to be literate have completed secondary school, compared to 15% who have completed primary school, 4% who have completed middle school, and another 4% who have completed Matric. Only 1% of the scheduled castes have graduated, and only a few are fortunate enough to have post-graduate degrees. The survey also supports scheduled caste communities' and leaders' claims that they are at the bottom of all social development indices and that government development initiatives are biassed against them (Shah 2007).

Samuel and Nisar (2020) also concluded that Prejudice was directed against Christians, which was both a curse and a blessing. Muslims have been disproportionately disadvantaged by educational prejudice in government schools, although Christians have had little trouble gaining admission. This is due to Christians' ability to hoard the opportunities provided to them initially through unequal access to government

employment. Corruption, according to slum dwellers, is a key impediment to getting an education and finding work. The inequality with scheduled castes is a unique kind of disparity, of ideas and values dependent on the purity- pollution which is the basic to the hierarchy in caste structure (Dumont 1980). It is widely acknowledged that boosting female school attendance is beneficial to both individuals and societies. It's also a question of social justice. Girls are still disadvantaged in several nations in terms of school access and experiences and are more likely to drop out than boys. When girls are further disadvantaged, such as by their social or economic circumstances, they become progressively marginalized (Bhagavatheeswaran 2016). By developing human capital, education adds to a country's economic growth (Abuya et al. 2014). Higher-education countries are more likely to be politically stable, have better chances of survival, and achieve significant health gains (USAID 2008).

A study by Hossain (2016) on Dalits in Bangladesh found that all of the Dalit children, who were interviewed, had been humiliated by their senior students, either directly or indirectly. Dalit children are usually placed in groups of 32 at educational facilities. They are treated with scorn and humiliation by students from mainstream culture, making them feel uncomfortable and powerless, and they develop a sense of inferiority. Because of their uneasiness and inferiority complex, they typically enroll in school as a group. When his or her colony's buddies drop out, a solitary child rarely takes a class and left school. Due to caste-based shame and a fear of physical insecurity, they are unable to attend school alone. As a

result, when people are alone, they tend to give up. The ability of Dalit parents to invest in their children's education is constrained by their income poverty as a result of being restricted to menial jobs. Low parental education and employment status have a negative impact on children's education since both factors affect income and awareness levels. Parents' inadequate educational skills can increase the cost of their children's education because youngsters receive minimal help from their parents with schoolwork, necessitating expensive private tuition. Unsuitable family conditions make it difficult or impossible for them to study at home, and cultural factors worsen the situation. Girl children's education is impacted by a lack of basic amenities, child marriage, insecurity, fear of defamation, gender bias, and traditional ideas about their destiny (Hossain 2016).

Education was once needed for cultural, religious, and social development. Schultz and Becker's major work on the concept of human capital investment in the 1960s showed that higher education is a prerequisite for economic growth and that no country can achieve meaningful economic growth if its population is not educated. Most of them are illiterate. The rapid development of East Asian countries can be attributed to their excellent education systems (Khan and Mahmood 1997). Existing research concentrates on religious minorities' slum disparities rather than comparing the minority to the mainstream to get a clear picture of the issue. The limited studies on religious minorities in slums that have been published demonstrate a complicated interplay of social, economic, cultural, and physical difficulties that prevent religious minority children

from attending school in developing countries (Hossain 2016; Husain 2005). Over the last half-century, Pakistan has consistently neglected education. Education has never received adequate funding. The alignment of the education system and corruption is on the rise. Most governments, both at the provincial and federal levels, have failed to appoint literacy ministers. Education has never been considered important for the development of our country. Infrastructure, such as dams, bridges, roads, highways, industries, and agriculture, was given priority, but education and other social sectors were never considered (Khan and Mahmood 1997).

The major problem behind labor bounded and landless peasant is the quality of education they get. Now a day's education plays a major role to boost up towards the best way of living life and improving socio-economic position in the society, in the Kolhi community there is a lack of both things. "Education is viewed as a vehicle to lift socially and economically for marginalized communities (Hossain 2016) and "to acquire the way to take an interest in their communities" (OHCHR, as cited in UNICEF and UNESCO 2007). "Having a good education helps people to think and to live a better life in the society; it improves not only an individual but also their whole community through new ideas and social skills" (Al-Shuaibi 2014). General knowledge and social communication skills are more beneficial for everyone to manage livelihood (Groot 2007). Scheduled Castes (SCs) are considered as perhaps the most educationally backward in India. In higher education, with just 2% as non-technical graduates and

above, and just 0.7% with technical higher education. However, the general portrayal is poor yet (Vasavi 2012: Parul 2014). An appropriate education and training framework indeed adds to the stable political, financial, and social status of a state, like Pakistan; an under developing country with a feeble economy and unstable political conditions (Ashraf and Ismat 2016). In Pakistan, the nature of primary and secondary education has a declining pattern. It is understood that science education is arriving at the lowest ebb, should be improved desperately (Memon 2007).

The current education delivery system is not addressing the requirements and goals of the society as such for the betterment of modern ways, and it is a challenge to the territories and regions for the 21st century. Moreover, preceding devolution, the policy, and planning have been embraced by the focal and provincial governments without considering the ground real factors and without the participation of the local community (Shah 2003). The Scheduled castes have been viewed as the most vulnerable constituents of the Indian social structure. The Scheduled positions throughout the nation occupy the lowest ranking in the caste hierarchy and dealt with numerous issues, which are social, financial, political, and educational (Parul 2014). As Ravi Babu and Chandrasekarayy (2015) write that "they remain as the casualty regulated maltreatment. The majority of scheduled castes are having low education status, which is the foundation for backwardness with illiteracy, low income, landlessness, poverty, and so on. Despite governmental policy regarding minorities in society with

different educational and development programs, the situation with scheduled castes has not improved to the ideal level".

Hossain (2016) found in his studies that "a combination of various economic, social and cultural deprivations emerges out of their caste and plunge based stigmatization and discriminations exclude them from school". The disparity in educational opportunity is of different types: inequality by profession, inequality by basic resources, regional inequality, inequality by sex, inequality by caste, inequality by parental income, inequality by parental occupation, and so on. Despite numerous endeavors to change, the disparity with various faces persists in our educational system (Tilak 1979).

A study conducted by Karade (2008) revealed that "the backwardness of Dalit communities will be shown from their illiteracy, Joblessness and dependent on agriculture and social segregation". Those from the Kolhi, Bheel, or Meghwar families who have acquired an education and established themselves as teachers, lawyers, or social activists, have felt the discrimination, they suffered during their school, "Banned from sitting in the front rows and from drinking water from the common glass often made them feel the pangs of untouchability (Rehman 2015). This discrimination also prevailed in other developed countries like the US. According to Naik (2021) that "One out of three Dalit students, understudies reported being victimized of discrimination during their schooling in the US, two out of three Dalits reviewed detailed being

treated unfairly at their working environment in the US". A survey conducted in Bangladesh (2008) found that the "enrolment rate of Dalits in school was 10 percent and the dropout rate among enrolled Dalit children was 95 percent" (Chowdhury 2009, as cited in Hossain 2016: 2). The survey also found that the dropout ratio was high in class 2<sup>nd</sup> and 3<sup>rd</sup> (Islam et al. 2015: 12); they usually dropped out from early education that is more important. According to Pantawane "Dalit is not a caste; Dalit is a symbol of change and revolution" (As cited in Hossain 2016). A study by Jain (1981) on "Liberation of scheduled castes and Tribes: Some Suggestions" tracked down that the scheduled castes have little admittance to education which is the poor's most powerful weapon for self-headway.

Rural poverty, urban governance, school performance, bonded labor, and displacement are just some of the social policy topics we've collaborated on. Our most recent assignment took us on a whirlwind journey across the country, from Khyber to Karachi and from Badin to Gwadar. Understanding and documenting the various processes of social backwardness was part of this brief. Alternatively, systematic backwardness of individuals, families, and groups based on "social" characteristics such as caste, traditional occupation, kinship, race, religion, and lifestyle (Gazdar 2007). Several studies undertaken over the previous two decades have revealed a number of factors that influence school enrolment, but poverty has gotten little attention in this context. Because poor children are less likely to attend primary school and because the negative impacts of poverty are expected to be more pronounced in girls,

the possibility that the poor behave differently when determining whether or not to enroll their children in a primary school should be examined. The goal of this study is to look into the impact of poverty on primary school enrollment in Pakistan, as well as the gender disparities in enrollment when poverty is included (Arif et al. 1999).

Less developed communities have long attracted the interest of stratification researchers because of the clear differences in conditions between them and more industrialized countries. These may include less segregated class structures, educational systems that differ in their institutionalization, and lower levels of economic development and professional structures formed from weaker positions in the global system. Research in such a social environment can help in the development of new ideas. These cultural inequalities often do not provide opportunities for testing and expanding the theories that have emerged from research on industrialized countries (Buchmann and Hannum 2001). Low-caste children would still be at a disadvantage even if all teachers taught similarly in class since they require "more" academic assistance than their peers. The paper emphasizes the significance of accounting for the political economics of caste-based frameworks in which education is provided, even though the study's findings are limited due to its qualitative nature. Future research should focus on caste-based concerns within schools, among classmates, or among teachers, to fully understand how caste plays out on school grounds (Tamim and Tariq 2015). Scheduled castes, the lowest castes of Pakistani Hindus, are the worst victims of caste, descent, and occupation discrimination. They have a double disadvantage as Hindus (minorities in a Muslim-majority country) and members of Hindu society's lower castes. "Achoots" or "untouchables" are the terms used to describe them (Shah 2007).

Once upon a time, there was a list of major interest groups or "communities." This includes "deficiencies' (service castes) and "male" Muslim sheikhs in Punjab and NWFP, as well as "Christian" Christians in Punjab and Hindus of Bhil, Kolhi, and other listed castes in Sindh, who face religious discrimination. had to accept oppression based on caste. Traditional beggars, monks, and other nomadic groups represent another form of social exclusion from the mainstream economy and society. We have documented incidents of abuse of "lower caste" women by the Chauhra, Musli, Lachhi, and Scheduled Caste Hindu groups in Peshawar, Faisalabad, Quetta, and Sanghar, respectively. The perpetrators were all well-known, and there was a suspicion that they committed the murder knowing that the victims were socially and politically weak. In fact, these rapes were only the most extreme cases of sexual violence against the underprivileged (Gazdar 2007).

There has also been considerable research into how factors such as socioeconomic status, family structure, and material resources affect the educational outcomes of children in poor countries. Some studies have looked at how family and school characteristics interact to cause educational stratification, possibly as a result of focusing on the "family

vs. school" conflict described below. Rather, school-based drivers of educational classification have emerged relatively independently of those determining family foundations of educational stability, and such research studies school procedures or material inputs as they relate to acquisition and acquisition (Buchmann and Hannum 2001).

The life that Kolhi lives in villages of district Badin has hardly changed from what they have been for several centuries. Their eating habits, dressing style, cultural practices, language, and the superstitious ritualistic faith system is still much the same, although some modernizing effects in the form of transportation facilities and road have made some impact on their life, but, by and large, all modern facilities that people of other communities of Pakistan avail in other districts, are mere nominal in district Badin and Tharparkar especially for Kolhi community (Hasan 2009; Shah 2007). As Mehdi (2019) pointed out that, "Feudal and traditional moguls' system has ruined the fertile desert of Pakistan. [One should take responsibility] by providing all-important infrastructure and fulfillment of basic needs for the standard livelihood". Suneila Malik (1979) found that the formative measures have helped social mobility among the scheduled. But still a larger part of the Scheduled Castes can't profit to the furthest reaches from these developmental measures because of escape clauses in the execution system. Kirpal (1978) expressed that the scheduled castes are treated as outsiders and untouchables for quite a long time at a stretch. Caste is one such effective stratifying feature that can interact in a variety of ways with class, culture, ethnicity, religion, and language, ensuring the social isolation of stigmatized groups across generations and limiting their social mobility and educational prospects (Kabeer 2006; Drèze and Kingdon 2001; Kingdon 2009). And Kulkarni (1974) dissected the report that there is significant variation in the degree of education among the various networks of the scheduled castes and Scheduled Tribes themselves. In a study on Dalits and education, there are various reasons advanced to disclose the absence of admittance to schooling for Dalit children. One methodology centers around destitution; another looks at social factors where even when Dalit students figure out how they are discriminated in school (Hossain 2016). The structural isolation of low-caste children from schools reflects their social constraints, which are mirrored in teachers' academic judgments (Bourdieu 1986).

The caste system in India is a social hierarchy derived from the Varna system in the founding text of Hinduism. The eldest is Varna Brahman (priest), followed by Kashtari (soldier or administrator), Vaishya (merchant), and Shudra (slave) (servant). There are also Dalits whose traditional duties include removing human exclusion from the system. The Dalits, who make up about 200 million of India's 1.3 billion inhabitants, still face discrimination in the country. The upper castes are known as "untouchables" because they refuse to touch anything that comes in direct contact with them. Mahar, a special Dalit caste to which Ambedkar belonged, had to clean his tracks by sweeping around his rooms. Even in 1998, about 90% of those employed by the Indian government as

"sweepers" were Dalits, who removed human waste from bathrooms. In Ambedkar's time, there were massacres of lower castes at the hands of upper castes, as is the case in India today (Perrigo 2020).

In a 1927 speech of Ambedkar observed, "Untouchability is not an easy task." "It is the source of all our poverty and deprivation, and it is the cause of our current plight ... The fundamental injustice of the four-caste system must be eradicated." According to Shah (2007) due to a lack of equity, property ownership, and economic ventures, the scheduled castes rely on wage labour for their livelihood. The bulk of them live and work as agricultural workers in rural areas, with some working as daily wagers in nearby towns and cities. They are frequently forced to perform menial activities that are not performed by Muslims or upper-caste Hindus.

In Sindh where most Dalits live, they are not represented in Sindh Assembly by any independent and genuine Dalit leader, thus leading to further exclusion of Dalits from policymaking, and fund-allocation processes, and consequently their funds are then spent up by upper-caste Hindu minority members (The News 2012). Upper caste Hindus are elected on basis of Scheduled castes and do not speak for them at the national and provincial level thus creating discrimination for Dalits or lower communities (Rehman 2017). "There is no particular law in Pakistan to condemn discrimination against members of scheduled castes," says Rochi Ram, a senior lawyer and member of the Human Rights Commission of Pakistan (HRCP). "Caste isn't an issue but no one wants to

discuss castes in Pakistan." and Rochi additionally said that "Nobody has been selected as a judge or a magistrate from scheduled caste in the country" (as cited in Shah 2007: 7). In recent years Dalit community has made some progress at the national and provincial levels. Ms. Kirshna Kolhi is the first Dalit woman nominated as a senator by PPP since 2018 in a Muslim majority country [Pakistan] for the first time (Aljazeera 2018). Kirshna Kumari "had an intense childhood when she alongside her relatives and family members were held for a very long time as bonded labor in a private prison allegedly owned by the landlords of Kunri of Umerkot area" (Dawn 2018), Even she said that "In school we—the young ladies from the Kolhi community—were not allowed to drink water from a public glass. At the point when we were thirsty, we usually had a cup in our hands and have water poured in our grasp from high up." (As cited in PDSN and IDSN 2013). Her brother Veerji Kolhi, Senior member of the Human Rights Commission of Pakistan and special assistant of the Chief Minister of Sindh for Human Rights Department, Government of Sindh said in an interview with Daily Times that, being born in a labor bounded family, life was not less than a challenge as his whole community was politically, socially, and economically deprived. On one side, they were Hindus while on the other side they were under the dark shadow of feudalism. This was told by Veerji Kohli, a rights activist, and lawyer who belongs to a marginalized community of Thar. "He chooses not to give up to the standards set by feudal and challenged their rules as a huge number of individuals from Thar were taken as bounded labor. Their essential

rights were being disregarded," he said (Malik 2018). Furthermore, there is significant evidence that gender social and educational experiences, ranging from social attitudes for educated women to areas of gender discrimination and violence in schools, have a significant impact on women's participation and learning (Stromquist and Fischmann 2009). The levels of poverty in rural sections of the same country might differ considerably. The poorest areas in Pakistan, for example, are mostly in arid and desert zones, as well as ecologically sensitive steep hill-slope areas. Such locations are frequently secluded in every aspect. There are few non-farm job openings, and labour demand is mainly seasonal. Others among the poor live in locations with more natural resources but limited access to social services (education and healthcare) and infrastructure-electricity, irrigation, information and technical assistance, transport, and market centres- (Chaudhry et al. 2006).

The Kolhi people do not have adequate drinking water. Every day the women have to collect water and firewood from a distant source of well or hand pump (Hussain 2014). Dalit women are minimized due to their frail financial status, low schooling levels, and their area in cruel states of work like labor bounded, cotton picking, and working in bricks factories (PDSN and IDSN 2013). "If those feudal and landlords liked someone's girl (Dalit Communities), they would kidnap them, rape and convert them to Islam. How might you change somebody's religion like that? In any case, this is the thing that occurred there. It's not right but it happened openly," Said Veeru Kolhi, (Ghani 2017). This forced conversion of Religion to Islam is

also a restriction for girls to access education. Due to forced conversion, their parents do not allow them to go to school. In rural areas, local politics was organized with caste rather than class or ideology, according to Alavi's (2001) research on patron-client interaction, which resulted in unequal access to public resources. According to Qadeer's (2012) research, caste was associated with property ownership and professionalism, which resulted in many losses. The 'class/caste linkage', according to Gazdar and Mallah (2011), was critical in maintaining class dominance and hierarchical rank between the landowning and nonlandowning castes. According to their research, the government system of allocating property to non-cultivators in the 1970s did not achieve the expected results due to the 'authority of the original proprietors,' the high-castes, to eject anyone from the village. In districts and locations where the scheduled caste community predominates, education is in disarray. The literacy and enrollment rates in these districts are quite low. Scheduled caste children are unable to attend school due to a variety of factors. The dearth of schools in the villages, the inadequate infrastructure of existing schools, the shortage of teachers, discrimination, and, most importantly, the children's poor health are all issues that need to be addressed. Malnutrition has been demonstrated to have a deleterious effect on student participation and academic achievement (Shah 2007).

A couple of studies talked about those inequalities that exist for marginalized minority communities (scheduled castes). This leads minorities to be excluded from the social world, yet these examinations

don't talk about the disparities in promising circumstances between various religious groups. Accordingly, factors driving and contributing towards imbalances and setting out freedom holes need critical. To improve the existence of scheduled castes, "that creating these inequalities and the opportunities available to these marginalized, should be studied in more prominent detail and need to be clearly understood" (Samuel and Nisar 2020). In a study by Tamim and Tariq (2015), as a result, social classes outside of school infiltrate classrooms under the guise of educational assessment (ibid.). Due to the dominance of upper caste social networks, lower castes have lost the ability to hold public school teachers accountable. Only when the interests of lower and upper caste families clash, when their children attend the same government school, as in Chakwal, the schools are held accountable, leaving the children of both lower and upper caste Benefits. Another study findings reveal that schools discriminate against and label those who are stigmatized in society as failures, leading them to believe that they are 'naturally' unfit for education and, as a result,' removed' from school (Bourdieu and Passeron, 1977). The majority of teachers in the district where the study was conducted were harsh with scheduled caste youngsters, using disparaging language and humiliating them with corporal punishments, according to the children and their parents. Teachers were commonly accused of forcing students to sit in the back seats, cleaning and washing classrooms and washrooms, and making fun of students' attire and shoes According to the survey's findings, the vast majority of respondents (76 percent) said their children face bias

in school as a result of their caste. Teachers are sometimes accused of addressing students as "O Bagri, Bheel" instead of their given names. The requirement that all students receive free books and other materials discriminates against children. Students from Scheduled Castes are not eligible for Zakat stipends, which are only available to Muslim students.

Ambedkar's first book, "Who Were the Shudras?" (1946), investigates the concept that the Indo-Aryans' social organization was based on the Chaturvarnya philosophy, which divides society into four classes: Brahmins or priests, Kshatriyas or soldiers, Vaishyas or traders, and Shudras or menials.' Chaturvarnya employed the concept of "graded inequality," according to Dr. Ambedkar, to set the terms of connected living between classes. According to this system, "the Shudra is not only placed at the bottom of the gradation but he is also subjected to several ignominies and impairments so that he does not ascend above the state designated by law." The word 'Shudra,' according to Dr. Ambedkar, is a tribal name, not a derivation word that means 'one overcome by melancholy,' as some old publications indicate. "The Brahmanic writers outperform everyone in the art of producing phony etymologies," he writes. Dr. Ambedkar further shows that the Shudras were Kshatriyas and that some of them were rulers in ancient Aryan communities. He believes the Shudras were 'degraded' and consigned to the fourth Varna as a result of a 'violent struggle' between Shudra kings and Brahmins. Even if India gained independence, Dalits, according to Ambedkar, would stay at the bottom of society unless caste was abolished by everybody. "Whether the

Congress is fighting for freedom or not is incidental to the question of whose freedom the Congress is fighting for," he said.

Chapter No: 3

RESEARCH METHODOLOGY

This chapter explains the detailed research methodology that was employed to identify the factors that bar the education of scheduled castes as well as to address the topic "Education of Kolhi community: opportunities and challenges" through a systematic and structured way. Data were collected from the students of the Kolhi community who were studying in 6<sup>th</sup> to 10<sup>th</sup> classes. Moreover, the study aims to identify experiences and challenges faced by the Kolhi community's students due to their lower castes. The quantitative method for data collection was used through primary sources by doing fieldwork. Furthermore, the details of research design, research population, sampling size, methods for data collection, research strategy, and type of data analysis are described in this chapter.

#### 3.1 Research Design

Quantitative research methods were used to identify problems faced by Kolhi communities' youth to acquire proper education in Sindh, especially in district Badin. In this type of method, the researcher used a questionnaire made up of 17 statements using the Likert scale. Moreover, the researcher collected data through face-to-face interaction with the students who were studying in classes 6 to 10. As respondents were not able to fill the questionnaire by themselves, therefore, the interviews were taken from them based on the statements.

# 3.2 Locale of the study

The locale for this study was district Badin, Sindh. The reason behind the selection of this area was the huge population of the Kolhi community residing in the region.

#### 3.3 Target Population

The target population was the Kolhi community in district Badin. The data was collected from different villages. The study population was further limited to school students, from classes 6 to 10.

# 3.4 Sampling techniques

The researcher used random sampling technique to collect data. The reason behind the selection of this technique was, to represent the whole population of the community. As most of the people were living in small groups of 8 to 10 houses in a village and the ratio of educated youth was very low in each village that's why researcher used to approach randomly to the respondents.

### 3.5 Sample Size

The study sample size is based on 180 students from different villages of district Badin, who were selected by getting a random sample of class  $6^{th}$ ,  $7^{th}$ ,  $8^{th}$ ,  $9^{th}$ , and  $10^{th}$ .

#### 3.6 Tools for Data Collection

In this research closed-ended questionnaires were designed to collect the data. The questionnaire was divided into two sections, one section was of

demographic information of the respondents, and the second section was of the statement based on the objective of the study.

# 3.7 Techniques for Data Collection

In this study, closed-ended questionnaires were filled in face-to-face interaction, as the respondents could not fill out the questionnaires on their own.

# 3.8 Tools for Data Analysis

The data were quantitative. Therefore, Statistical Package for Social Sciences (SPSS) was used to analyze the data. The analysis is univariate and presented in a simple frequency distribution.

#### 3.9 Ethical Concern

The maintenance of ethical standards is important for any researcher while conducting a research study. The topic was much problematic because of traditional communities, where young children were not allowed to talk in front of their parents and were held in high regard. While interviews it was assured to all respondents that the information, they have provided will be kept confidential. and the interviews were conducted in a friendly manner. Most of the respondents were known to the researcher, so it was easy to get real and reliable data and no personal questions were asked that can hurt the respondents.

Chapter No: 4

FINDING AND RESULTS

This chapter reports the results and findings of the study. A simple questionnaire was made of two parts one is about demographic information and another was about simple statements related to their educational barrier that limits the majority of the community access to education. Moreover, the chapter defines the findings in the form of tables and well description of them. Frequencies and percentages of respondents are shown in each table.

Table 4.1. Age of the respondents

Categories	Frequency	Percent
10-20	164	91.1
21-30	15	8.3
31-40	1	.6
Total	180	100.0

Table No: 4.1 shows that 91.1% of the respondents are between 10-20 years old and 8.3% are between 21-30 years, while only 0.6 % represents the respondents between the ages 31 and 40 years. The respondents who aged above 20 years had left the school during 9<sup>th</sup> and 10<sup>th</sup>. However, the researcher focused on teenagers who were currently enrolled in educational institutions.

Table 4.2. Education level of the respondent

Categories	Frequency	Percent
6 <sup>th</sup>	34	18.9
$7^{ m th}$	25	13.9
8 <sup>th</sup>	28	15.6
9 <sup>th</sup>	51	28.3
10 <sup>th</sup>	42	23.3
Total	180	100.0

According to table 4.2, 18.9% of the respondents are from the 6th class, 13.9% from the 7th class, 15.6% from the 8th class, 28.3 % from the 9th class, and 23.3 % from the 10th grade with a majority of the responders in the 9th and 10th grades.

Table 4.3. Occupation or Profession of the respondent

Categories	Frequency	Percent
Student and Labor Worker	90	50.0
Farmer	23	12.8
Tailor Master	7	3.9
Student	52	28.9
Student and Shopkeeper	8	4.4
Total	180	100.0

Table No. 4.3 demonstrates that 50% of the respondents are laborers, 12.8 % are formers, and 3.9% are tailor masters. While 28.9% of those polled are students, 4.4% of those polled are students who also work as shopkeepers. The majority of the respondents work as laborers after school, according to this data. The rest of the responders, on the other hand, undertake a different kind of work during their education.

Table 4.4. The school is quite away from your home.

Categories	Frequency	Percent
Agree	74	41.1
Strongly Agree	102	56.7
Disagree	4	2.2
Total	180	100.0

According to Table No. 4.4, 41.1% of respondents agree and 56.7% strongly agree that the school is far away from their house, while just 2.2% of respondents disagree with the statement made by 180 respondents. According to the table, the majority of respondents do not have schools within walking distance of their residences.

Table 4.5. Your parents want you to get a proper education.

Categories	Frequency	Percent
Agree	113	62.8
Strongly Agree	5	2.8
Neutral	21	11.7
Disagree	40	22.2
Strongly Disagree	1	.6
Total	180	100.0

Table No: 4.5 explains that 62.8% of the respondents agree that their parents want them to get proper education, and only 2.8% of the respondents strongly agree to the above statement. While 11.7% of the respondents neither agree nor disagree, 22.2% of the respondents disagree and 0.6% of the respondents strongly disagree with the given statement. This table concludes that the majority of parents of the respondents want to get educated to their children. However, 22.2% of respondents argued that their parents do not want them to go to school, because they want their children to earn money to manage their expenses by own due to early marriages in the Kolhi community as a part of their culture.

Table 4.6. Your family has enough money to support your education.

Categories	Frequency	Percent
Agree	36	20.0
Strongly Agree	24	13.3
Neutral	3	1.7
Disagree	110	61.1
Strongly Disagree	7	3.9
Total	180	100.0

Table No: 4.6 shows that 20% of the respondents agree, and 13.3% of the respondents strongly agree that their family has enough money to support them in education and only 1.7% of the respondents stayed neutral. While 61.1% of the respondents disagree and 3.9% of the respondents strongly disagree with the above statement. In conclusion, it is clear that the majority of the respondents disagreed that they do not have support from family for their education expenditures.

Table 4.7. You work as labor after school time.

Categories	Frequency	Percent
Agree	34	18.9
Strongly Agree	144	80.0
Neutral	1	.6
Disagree	1	.6
Total	180	100.0

According to Table No. 4.7, 18.9% of respondents agree and 80% of respondents strongly agree that they work as labor after school hours. Only 0.6 percent of respondents remained indifferent, while 0.6 percent disagrees with the statement. The majority of respondents strongly agree that they work as labor after school hours, according to this table.

Table 4.8. Your family has enough money to buy books and copies.

Categories	Frequency	Percent
Agree	32	17.8
Strongly Agree	26	14.4
Neutral	2	1.1
Disagree	108	60.0
Strongly Disagree	12	6.7
Total	180	100.0

Table No: 4.8 explains that 17.8% of the respondents agree that they have a family to buy books and copies and 14.4% of the respondents strongly agree while 1.1% of the respondents stayed neutral. 60% of the respondents disagree and 6.7% of the respondents strongly disagree that they do not have family support to buy books and copies. Thus, this table concludes that the majority of the respondent's families do not have money to buy books and copies for their children.

Table 4.9. Your parents have financial support to send you for secondary education.

Categories	Frequency	Percent
Agree	7	3.9
Strongly Agree	2	1.1
Neutral	1	.6
Disagree	75	41.7
Strongly Disagree	95	52.8
Total	180	100.0

Table No. 4.9 shows that 3.9 percent of respondents agree, and 1.1 percent strongly agree that they have financial support from their parents to pursue secondary education, while 0.6 percent remained neutral, 41.7 percent disagree, and 52.8 percent strongly disagree that they do not have financial support from their parents. According to this data, the majority of respondents do not have parental support to pursue higher education as they work on the land and barely make enough money to eat and shelter.

Table 4.10. Your parents want that you should work with them on the land.

Categories	Frequency	Percent
Agree	40	22.2
Strongly Agree	131	72.8
Neutral	6	3.3
Disagree	3	1.7
Total	180	100.0

Table No: 4.10 explains that 22.2% of the participants agree and 72.8% strongly agree that their parents want them to work with them on the land while 3.3% of the participants stayed neutral and 1.7% disagree with the given statement. This data implies that many of the respondents' parents want their kids to work on the farm with them after school or instead of going to school.

Table 4.11. Your parents' expectation after not getting any job.

Categories	Frequency	Percent
Agree	122	67.8
Strongly Agree	19	10.6
Neutral	32	17.8
Disagree	7	3.9
Total	180	100.0

Table No: 4.11 depicts that 67.8% of the respondents agree that their parents expect nothing about further studies after matric or intermediate, 10.6% of the respondents strongly agree with the statement while 17.8% of the respondents stayed neutral and 3.9% of the respondents disagree to the given statement. According to this chart, the majority of respondents believe that their parents expect to acquire a job after finishing matric or intermediate school and there will be no need to continue their education further. And the main reason for not sending them for further studies is that their parents do not have enough money to send their children for higher education. Their parents work all day on the land, and barely earn money for survival. The ratio of unemployment is already at the peak for the university level students in Pakistan, therefore their parents expect that their son earns money from labor work or any small business.

Table 4.12. Due to a lack of job opportunities, you prefer to work on agricultural land.

Categories	Frequency	Percent
Agree	112	62.2
Strongly Agree	39	21.7
Neutral	24	13.3
Disagree	5	2.8
Total	180	100.0

Table No. 4.12 represents that 62.1 percent of respondents agree that they prefer to work on the land after being unemployed, 21.7 percent strongly agree, 13.3 percent remained neutral, and 2.8 percent disagree with the above statement. This table concludes that the majority of the respondents prefer to work on agricultural land due to a lack of job opportunities after graduation.

Table 4.13. Your family migrates from one place to another for seasonal earning.

Categories	Frequency	Percent
Agree	63	35.0
Strongly Agree	41	22.8
Neutral	1	.6
Disagree	75	41.7
Total	180	100.0

Table No: 4.13 explains that 35% of the respondents agree and 22.8% of the respondents strongly agree that their family migrates from one place to another for seasonal earning. While 0.6% of the respondents stayed neutral and 41.7% of the respondents disagree with the given statement about their family migration. According to data collected, the majority of the respondents agree and strongly agree that their family migrates for seasonal earning in a different time as Kolhi community people usually migrate with families for the seasonal earning from agriculture land, they do migrate where they find a fertile land and meet with the owner of that land for cultivation of land for just livelihood. All of the agriculture's earnings and benefits will go to the landlord, and peasants will be lucky to receive enough money to eat on a daily basis.

Table 4.14. You cannot attend the school continuously, due to seasonal migration.

Categories	Frequency	Percent
Agree	53	29.4
Strongly Agree	51	28.3
Neutral	1	.6
Disagree	75	41.7
Total	180	100.0

Table No: 4.14 shows that 29.4% of the respondents agree and 28.3% of the respondents strongly agree that pupils cannot attend school in the neighborhood on a continuous basis due to seasonal migration. While 0.6% of the respondents stayed neutral and 41.7% of the respondents disagree with the statement given above. This table concludes that many of the respondents agree and strongly agree that they cannot attend school continuously due to their parents' seasonal migration. As the children are admitted to the school where they are living since their birth, but then their parents suddenly decide to migrate somewhere, away from the place of permanent living. Their parents also want that their children should work with them on farming as they can earn more money for livelihood.

Table 4.15. Do you think that your parents do your marriage during your studies at the school level?

Categories	Frequency	Percent
Agree	32	17.8
Strongly Agree	128	71.1
Neutral	17	9.4
Disagree	3	1.7
Total	180	100.0

Table No. 4.15 represents that 17.8% of respondents agree and 71.1 % strongly agree that their parents marry their children while they study at the school level, whereas 9.4 % of respondents stayed neutral and 1.7 % of respondents disagree with the above statement concerning marriage at school level. According to the results in this table, most students believe that their parents do marry while they are still in school. Many of those polled were already married with children under the age of five, and many more had children while in school up until matriculation.

Table 4.16. Your teachers discriminate against you based on your religion.

Categories	Frequency	Percent
Agree	64	35.6
Strongly Agree	2	1.1
Neutral	28	15.6
Disagree	85	47.2
Strongly Disagree	1	.6
Total	180	100.0

Table No: 4.16 shows that 35.6% of the respondents agree and 1.1% of the respondents strongly agree that the teacher discriminate with them on basis of their religion, while 15.6% of the respondents stayed neutral, 47.2% of the respondents disagree and 0.6% of the respondents strongly disagree to the given statement. According to the data in this table, the vast majority of students believe that their teachers do not treat them unfairly because of their religious background. However, 35.6% of those polled said they had experienced discrimination in the classroom from their teachers. The majority of those polled have no idea how teachers have a racial bias against them.

Table 4.17. Your teachers do not respond to you then they respond to Muslim fellows.

Categories	Frequency	Percent
Agree	72	40.0
Strongly Agree	54	30.0
Neutral	12	6.7
Disagree	42	23.3
Total	180	100.0

Table No: 4.17 describes that 40% of the respondents agree and 30% of the respondents strongly agree with the statement that their teachers do not treat them equally, while 6.7% of the respondents stayed neutral and 23.3% of the respondents disagree with the statement. According to this table, the majority of the respondents feel that their teachers do not respond to them equally through religious and/or occupational discrimination.

Table 4.18. You feel discrimination from your Muslim fellows in class.

Categories	Frequency	Percent
Agree	58	32.2
Strongly Agree	15	8.3
Neutral	40	22.2
Disagree	67	37.2
Total	180	100.0

Table No: 4.18 shows that 32.2% of the respondents agree, and 8.3% of the respondents strongly agree that they feel discrimination from their Muslim fellows, while 22.2% of the respondents stayed neutral, and 37.2% of the respondents disagree with the statement that they do not feel any type of discrimination from their Muslim fellows. This table concludes that the majority of the respondents disagreed that they do not feel any discrimination from their Muslim fellows. However, according to the percentage of agreed and strongly agreed students, 32.2% and 8.2% respondents agreed and strongly agreed that they have felt discrimination from their Muslim fellows.

Table 4.19. Your glass and plates are separate at the school canteen.

Categories	Frequency	Percent
Agree	8	4.4
Strongly Agree	171	95.0
Disagree	1	.6
Total	180	100.0

Table No: 4.19 describes that 4.4% of the respondents agree and 95% of the respondents strongly agree that there is separate glass and plates at the school canteen. While only 0.6% of the respondents disagree with the given statement above. This table concludes that the majority of the respondents strongly agree that their glass and plates are separate at the school canteen if some canteens do not have separate pots for them to eat then they cannot eat at that canteen.

Table 4.20. You are allowed to eat at the same food points in school.

Categories	Frequency	Percent
Agree	136	75.6
Strongly Agree	8	4.4
Neutral	18	10.0
Disagree	18	10.0
Total	180	100.0

Table No: 4.20 shows that 75.6% of the respondents agree that they can eat from the same food point in school and 4.4% of the respondents strongly agree, while 10% of the respondents stayed neutral and 10% of the respondents disagree with the statement. This table concludes the majority of the respondents agreed that they can eat at the same food points in schools.

# Chapter No: 5

# DISCUSSION, CONCLUSION, AND RECOMMENDATION

#### 5.1 Discussion

This chapter concludes the major findings of the study and discusses the findings by giving examples from some studies. An ethnographic study by Hussain (2014) pointed out that the Kolhi community is one of the lists of lower castes (scheduled castes) who are less educated and less economically developed, as compared to Meghwar, Bagri, Valimiki, and Oad in Sindh. Kolhi, also known as scheduled castes, landless peasants, and marginalized communities, the study findings reveal that the majority of the community works under the landlords in agriculture, as a former on the lands of Muslim landlords. However, this study aims to explore the reasons behind the lack of access among children of the Kolhi community to education and to learn from their experiences and what challenges they face when going to school. They are backward in the era of education and are discriminated against in every field of life due to their physical appearance, work, and caste. They are less aware of new modern technologies and their rights in society. They are humiliated and bound by Muslim landlords to work on their lands. The people of the Kolhi community have also developed a sense in their minds that they are made for it, to work on the land and to do labour work. Due to this sense, they don't even know that they are discriminated against, and they look like others. They feel happy to work on the lands of Muslim landlords for some money because they don't have any skills or knowledge to work on their own. However, some people from the community have educated themselves and their family members as like Veerji Kolhi and his sister

Kirshna Kumari, they are both politician and now working their own community. They have said that they faced discrimination in their early education and their family has been slaved under Muslim landlords. Kirshna Kumari "had an intense childhood when she, alongside her relatives and family members, was held for a very long time as bonded labour in a private prison allegedly owned by the landlords of Kunri in the Umerkot area" (Dawn 2018). Even she said that "In school we—the young ladies from the Kolhi community—were not allowed to drink water from a public glass. At the point when we were thirsty, we usually had a cup in our hands and had water poured into our grasp from high up. " (as cited in PDSN and IDSN 2013). The NGOs that are working in the rural areas of the districts of Tharparkar and Badin have also found that the feudals didn't want the lower community to progress in education.

From the findings of the study, it was found that many of the respondents strongly agreed that the school is quite away from their homes, which makes it a hurdle to attend school. Their family resides too far away from the city or village where the school is located. Therefore, if they want to go to school, then they should have to pay fares to Raksha, otherwise they have to walk a lot. They claimed that, as a result of this, many students cannot afford to travel to school, which is located far from their home. Because the majority of the respondents' parents are farmers and they live in rural areas, a long walk distance from the city, where they struggle to make ends meet on a daily basis, their parents do not consider investing in their children's education. Several studies undertaken over the previous

two decades have revealed a number of factors that influence school enrolment, but poverty has gotten little attention in this context. Because poor children are less likely to attend primary school and because the negative impacts of poverty are expected to be more pronounced in girls, the possibility that the poor behave differently when determining whether or not to enroll their children in a primary school should be examined. The goal of this study is to look into the impact of poverty on primary school enrollment in Pakistan, as well as the gender disparities in enrollment when poverty is included (Arif et al. 1999). As Chowdhury (2009) defines that historically Dalits are found victims of social exclusion and discrimination that limit them to access education. Therefore, the current research has tried to answer the question; what are the major reasons behind labor bounded and landless peasants of the Kolhi community getting proper education? The Dalits also called scheduled castes or untouchables "have experienced predictable denial to access to education since the 1850s (Fraser 2010)

The study also found that most of the parents do not have interest in whether their children are going to school or not. They mostly think that their children should work with them and would do good labor work to earn money. Many of the respondents also agreed that their parents want them to work on farming land instead of going to school. However, majority of the respondents argued that their parents do not want them to go to school, because they want their children to earn money to manage their expenses by own as they can do early marriages, which is a part of

their culture in the Kolhi community. During field work of the study, the researcher found that students who were studying in schools have to do a lot of work with their parents to run whole family and to arrange food for them. Most of students was working in the land with their parents after school time. Its really hard for student to do work while attending school because they do not get time at home to study. They manage such daily routine time for their work. Due to caste-based shame and a fear of physical insecurity, they are unable to attend school alone. As a result, when people are alone, they tend to give up. The ability of Dalit parents to invest in their children's education is constrained by their income poverty as a result of being restricted to menial jobs. Low parental education and employment status have a negative impact on children's education since both factors affect income and awareness levels. Parents' inadequate educational skills can increase the cost of their children's education because youngsters receive minimal help from their parents with schoolwork, necessitating expensive private tuition. Unsuitable family conditions make it difficult or impossible for them to study at home, and cultural factors worsen the situation. Girl children's education is impacted by a lack of basic amenities, child marriage, insecurity, fear of defamation, gender bias, and traditional ideas about their destiny (Hossain 2016).

Moreover, many of the respondents strongly agreed to the statement that they got married during their studies at the school level and some of them at the primary level. Early marriage tradition is also a big barrier with the Kolhi community's students to access higher education. Due to early

marriages, they become parents when they are in class 9<sup>th</sup> or 10<sup>th</sup>, and at that time, they do not get financial support from their parents. They start working as labor or in any business to run the family. As in a study found that unsuitable family conditions make it difficult or impossible for them to study at home, and cultural factors worsen the situation. Girl children's education is impacted by a lack of basic amenities, child marriage, insecurity, fear of defamation, gender bias, and traditional ideas about their destiny (Hossain 2016).

The majority of the respondent strongly agreed that they face discrimination by their teachers in schools based on their lower caste. They claimed that teachers are less responsive to them as compared to other Muslim fellows and most of the respondents also strongly agreed that their glass for drinking and pots for eating are separate in schools for scheduled caste Hindus, from which they belong. Hence, this type of injustice on the basis of the caste system creates a lack of interest among students in going to school. As a study by Hossain (2016) on Dalits in Bangladesh found that all of the Dalit children, who were interviewed, had been humiliated by their senior students, either directly or indirectly. Dalit children are usually placed in groups of 32 at educational facilities. They are treated with scorn and humiliation by students from mainstream culture, making them feel uncomfortable and powerless, and they develop a sense of inferiority. Because of their uneasiness and inferiority complex, they typically enroll in school as a group. When his or her colony's buddies drop out, a solitary child rarely takes a class and left school. Due to castebased shame and a fear of physical insecurity, they are unable to attend school alone. As a result, when people are alone, they tend to give up. Another study also found that Scheduled caste students and parents have alleged that their peers and teachers make fun of them because of their poor clothing and appearance. Many of the students in this group wore ragged clothes and were barefoot. Unfortunately, rather than integrating and promoting excluded groups into mainstream society, the entire system appears to alienate them even further, and there appears to be no systematic effort to reduce prejudice and discrimination (Shah 2007).

Many of the students could not attend school due to seasonal migration. Students argued that their family migrates from on place to another for seasonal earning. There are around 2,400 villages with a high Dalit population in Tharparkar district. The ratio of upper caste Hindus in this district, like the rest of Sindh, is only 15% (Kolhi 2014). Although the soil in Tharparkar is exceptionally fertile, crop production is dependent on summer rainfall due to the fact that the majority of the region is desert. Rain rains from mid-June to mid-August every year. Drought has been a problem in Tharparkar for the past ten years, forcing half of the people to migrate to neighboring locations in search of food, as it is impossible to acquire water to drink, let alone plant crops, during droughts. The literacy rate is poor in lower communities, and most people who migrated to barrage areas during droughts look for any work if they can. As a result, their children's education is neglected, and students cannot attend school as they are somewhere on the other region, and they hesitate to go school

which is new for them to be enrolled, children usually started working as child labor. While their parents travel, children must leave their education in the middle. Hamero lives in the Badin district of Sindh, in the village of Bandho. He is the father of seven children, three of whom attend Bandho village school. His children, however, had to leave school when he travelled, and he enrolled them wherever he could find a place for them to stay throughout the migration. His children were denied admittance to a nearby school because they required a high school diploma, which he lacked. He did not take a school leaving certificate because their move was only temporary. His children did not attend school and were no longer interested in studying when he returned after four months of labor. their parents also do not think about school admission as they know after monsoon weather, they have to go back their own home and land, from where they migrated for seasonal earning and most children are pulled out of school after a few years anyway due to a lack of funds. Several families have also faced rape, forced abductions, and sexual abuse of their children while migrating. Landowners who give them with a way to generate money for their needs have been known to kidnap or rape their young and minor girls. Suicides and rapes have been reported, but no effective action has yet been taken.

The study found that the students who were interviewed with closed ended questions, were working on shops, hotel and as a child labor. Majority of the students do not attend schools due to family expenditure or family pressure to run small business like shop. Education was once needed for

cultural, religious, and social development. Schultz and Becker's major work on the concept of human capital investment in the 1960s showed that higher education is a prerequisite for economic growth and that no country can achieve meaningful economic growth if its population is not educated. Most of them are illiterate. The rapid development of East Asian countries can be attributed to their excellent education systems. Over the last half-century, Pakistan has consistently neglected education. Education has never received adequate funding. The alignment of the education system and corruption is on the rise. Most governments, both at the provincial and federal levels, have failed to appoint literacy ministers. Education has never been considered important for the development of our country. Infrastructure, such as dams, bridges, roads, highways, industries, and agriculture, was given priority, but education and other social sectors were never considered (Khan and Mahmood 1997).

# **5.2 Major Findings**

- The majority of the students agreed that the school was quite away from their homes, and they do not have financial support to go for higher education by spending. Whereas their parents also do not have a positive response about the education of their children.
- Early marriage is a community tradition in which most children are married by their parents while still in school, a tradition that has been passed down from generation to generation. Due to early

marriage, children leave the schools in matric and cannot continue further education owing to family expenditures, and then they start working as labor, farmer, tailor master, and Raksha driver to run family expenditures.

- The children from religious minority or marginalized community face discrimination by their teachers in schools based on their lower caste. Many of the students argued that they felt ashamed by their Muslim peer groups by calling inferior. There are separate glasses and pots for eating for Scheduled castes Hindus in almost every school in district Badin, which creates a lack of interest among Dalit students, to go to school.
- Seasonal migration is also affecting majority of children's education. Children, however, had to leave school when their parents travelled, wherever they could find a place for them to stay throughout the migration. If their parents used to admit their children to school then children are denied admittance to a nearby school because they required a high school diploma, which they lacked. They did not take a school leaving certificate because their move was only temporary, and children did not attend school and were no longer interested in studying when he returned after four months of labor.
- The main problem behind labor bounded and landless peasant is the quality of education they get. They are more victims of deprived and excluded from basic resources like education. They

mostly work as farmers for Muslim landlords and live as landless peasants. Kolhi community is a marginalized community, having less literacy rate, high poverty rate, and less socially developed community.

#### 5.3 Conclusion

The study aims to identify experiences and challenges of scheduled caste Kolhi students regarding their education in Sindh, Pakistan. The study found that the majority of the students were unable to attend school due to a lack of financial support, early marriages, and the unavailability of schools near their homes. Parents do not think to invest in their children's education. As Hussain (2014) explained that "Most of the untouchables (Dalit) communities are bounded in structured ethnic discrimination and socio-religious exploitation", Kolhi community is one of them. They want their children to work as a labor or former on the lands of Muslim landlords who are "Zamindar" in those rural areas. Some of the NGOs working specially on the education of these lower communities but still do not succeed in their goals, to eliminate poverty and low literacy rate among bounded communities. The government should do more better opportunities for these lower communities in the field of education, these communities progress in education will also help the country to GDP rate and will also lead the country with high literacy rate. As a study shows that Pakistan's literacy rate is 52 percent, with just 80 million individuals out of 140 million able to read and write, according to official data. The majority of the illiterate population in these places is made up of women, children,

religious minorities, low castes, bonded labourers, and other vulnerable and marginalized groups (Shah 2007).

The students of the Kolhi community have also felt discrimination in school, such as less response from teachers towards them due to lower caste, and separate glass and pot for eating at school canteens. Thus, discrimination leads to a lack of interest among students to go to school. The Kolhi people do not have adequate drinking water. Every day the women have to collect water and firewood from a distant source of well or hand pump (Hussain 2014). Dalit women are minimized due to their frail financial status, low schooling levels, and their area in cruel states of work like labor bounded, cotton picking, and working in bricks factories (PDSN and IDSN 2013). Unfortunately, rather than integrating and promoting excluded groups into mainstream society, the entire system appears to alienate them even further, and there appears to be no systematic effort to reduce prejudice and discrimination. However, the government and development organizations must work together to enable community economic empowerment in order to reduce migration and save the futures of hundreds of children so that they can contribute to the country's progress and prosperity as adults. There is a stray need for mainstreaming the Kolhi community in education without discrimination.

#### 5.4 Recommendations

• The government needs to operate schools near homes of these scheduled castes and socially excluded communities, to ensure

Education for All. Moreover, the Government should pay them all the educational expenditures to bridge the educational gap between the communities.

- There is a need for these communities to be aware of the importance of education, and for that, different organizations should work for the awareness of these people from lower communities.
- The people who belong to these lower communities should play a
  vital role to boost up literacy rate and socio-economic condition of
  their respective communities. As they can live a better life beyond
  bounded in structural discrimination.

## 5.5 Limitations of the study

The present study has some limitations, when the researcher went to villages for fieldwork, most of the people asked that what will be the benefit for us. Secondly, respondents could not fill the questionnaire by themselves therefore researcher did face-to-face interactions with them. The study was further limited to students who were studying in classes 6 to 9, and the sample size was limited to 180 respondents.

### **REFERENCES**

- Abuya, Benta A., Elijah O. Onsomu, and DaKysha Moore. 2014. "Determinants of educational exclusion: Poor urban girls' experiences in- and out-of-school in Kenya." *Prospects* 44:381-394. https://doi.org/10.1007/s11125-014-9306-1
- Ahmed, Iqbal., Kahil U. Rehman., Asghar Ali., Itbar Khan, and Fazal A. Khan. 2014. "Critical Analysis of the Problems of Education in Pakistan: Possible Solutions." *International Journal of Evaluation and Research in Education (IJERE)* 3(2):79-84.
- Alavi, Hamza. 2001. "The Two Biraderis: kinship in rural West Punjab, in:T. Madan (ed) Muslim Communities of South Asia: culture, society and power." New Delhi, Manohar.
- Aljazeera. 2018. "Krishna Kholi: Pakistan's first Hindu Dalit senator."

  Retrieved on June 21, 2021.

  <a href="https://www.aljazeera.com/news/2018/3/4/krishna-kholi-pakistans-first-hindu-dalit-senator">https://www.aljazeera.com/news/2018/3/4/krishna-kholi-pakistans-first-hindu-dalit-senator</a>
- Al-Shuaibi, Abdulghani. 2014. "The Importance of Education." *Qatar Chronicle*. Retrieved from <a href="https://www.researchgate.net/publication/260075970\_The\_Importance of Education">https://www.researchgate.net/publication/260075970\_The\_Importance of Education</a>
- Ambedkar, Bhimrao R. 1947. Who Were the Shudras?. Thacker and Co.'s editions.

- Arif, Ghulam M., Najam U. Saqib., Ghulam M. Zahid, and Aliya H. Khan.

  1999. "Poverty, Gender, and Primary School Enrolment in
  Pakistan." *The Pakistan Development Review* 38(4):979-992.

  <a href="https://www.jstor.org/stable/41260217">https://www.jstor.org/stable/41260217</a>
- Ashraf, Muhammad A. and Hafiza I. Ismat. 2016. "Education and Development of Pakistan: A Study of Current Situation of Education and Literacy in Pakistan". *US-China Education Review* 6(11):647-654.
- Bates, Timothy M. 1998. *Race, self-employment, and upward mobility: An elusive American dream*. Johns Hopkins University Press.

  <a href="https://books.google.com/books?hl=en&lr=&id=1YnUn3hmauEC">https://books.google.com/books?hl=en&lr=&id=1YnUn3hmauEC</a>
  <a href="mailto:&oi=fnd&pg=PP11&ots=OQ8iU4\_iwN&sig=MGrvaaO9QTV9gf">https://books.google.com/books?hl=en&lr=&id=1YnUn3hmauEC</a>
  <a href="mailto:&oi=fnd&pg=PP11&ots=OQ8iU4\_iwN&sig=MGrvaaO9QTV9gf">https://books.google.com/books?hl=en&lr=&id=1YnUn3hmauEC</a>
  <a href="mailto:&oi=fnd&pg=PP11&ots=OQ8iU4\_iwN&sig=MGrvaaO9QTV9gf">https://books.google.com/books?hl=en&lr=&id=1YnUn3hmauEC</a>
  <a href="mailto:&oi=fnd&pg=PP11&ots=OQ8iU4\_iwN&sig=MGrvaaO9QTV9gf">https://books.google.com/books?hl=en&lr=&id=1YnUn3hmauEC</a>
  <a href="mailto:KRERCsx3p0HLw">KRERCsx3p0HLw</a>
  - tlteille, Andr . 1981. *The Backward Classes and the New Social Order*.

    Oxford University Press New Delhi.
- Bhagavatheeswaran, Lalitha., Sapna Nair., Hollie Stone., Shajy Isac.,
  Tejaswini Hiremath., Raghavendra T., Kumar Vadde., Mahesh
  Doddamane., H.S. Srikantamurthy., Lori Heise., Charlotte Watts.,
  Michele Schweisfurth., Parinita Bhattacharjee, and Tara S. Beattie.
  2016. "The barriers and enablers to education among scheduled caste and scheduled tribe adolescent girls in northern Karnataka,
  South India: A qualitative study." *International Journal of*

- Educational Development 49:262–270. https://doi.org/10.1016/j.ijedudev.2016.04.004
- Bourdieu, Pierre and Jean-C. Passeron. 1977. "Reproduction in education, society, and culture." *Beverly Hills, CA: Sage*. http://refhub.elsevier.com/S0738-0593(15)00049-8/sbref0035
- Bourdieu, Pierre. 1986. *The Forms of Capital*. Handbook of Theory and Research and the Sociology of Education. Westport, CT: Greenwood.

https://www.socialcapitalgateway.org/sites/socialcapitalgatewa

- Buchmann, Claudia and Emily Hannum. 2001. "Education and Stratification in Developing Countries: A Review of Theories and Research." *Annual Review of Sociology* 27:77-102. <a href="https://www.jstor.org/stable/2678615">https://www.jstor.org/stable/2678615</a>
- Charsley, Simon R. and G. K. Karanth. 1998. *Challenging untouchability:*Dalit initiative and experience from Karnataka. SAGE

  Publications.
- Chaudhry, Imran S., Shahnawaz Malik, and Muhammad Ashraf. 2006.

  "Rural Poverty in Pakistan: Some Related Concepts, Issues and
  Empirical Analysis." *Pakistan Economic and Social Review*44(2):259-276. <a href="https://www.jstor.org/stable/25825297">https://www.jstor.org/stable/25825297</a>

- Chowdhury, Iftekhar U. 2009. "Caste-based Discrimination in South Asia:

  A Study of angladesh." New Delhi: Indian Institute of Dalit

  Studies. Working Paper Series III(7). <a href="http://idsn.org/wp-content/uploads/user\_folder/pdf/New\_files/Bangladesh/Caste-based\_Discrimination\_in\_Bangladesh\_IIDS\_working\_paper\_.pdf">http://idsn.org/wp-content/uploads/user\_folder/pdf/New\_files/Bangladesh/Caste-based\_Discrimination\_in\_Bangladesh\_IIDS\_working\_paper\_.pdf</a>
- Dawn, 2018. "In historic first, a Thari Hindu woman has been elected to the Senate". Retrieved on June 27, 2021. <a href="https://www.dawn.com/news/1393129/in-historic-first-a-thari-hindu-woman-has-been-elected-to-the-senate">hindu-woman-has-been-elected-to-the-senate</a>
- Deen, Sahab. 2015. "Determinants of Higher Education for Scheduled

  Castes in Uttar Pradesh: A Socio-Economic Analysis." Research

  Journal of Social Science & Management-RJSSM 5:212-217.

  Retrieved from

  <a href="https://www.researchgate.net/publication/290429250">https://www.researchgate.net/publication/290429250</a>
- Dove, Linda A. 1980. "The Role of the Community School in Rural Transformation in Developing Countries." *Comparative Education* 16(1):67-79. Retrieved on June 15, 2021. <a href="http://www.jstor.org/stable/3098553/">http://www.jstor.org/stable/3098553/</a>
- Drèze, Jean and Geeta G. Kingdon. 2001. "School Participation in Rural India." *The Suntory and Toyota International Centres for Economics and Related Disciplines London School of Economics Houghton Street*, Available at SSRN: <a href="https://ssrn.com/abstract=1126987">https://ssrn.com/abstract=1126987</a>

- Dumont, Louis. 1980. Homo Hierarchicus: The Caste System and Its Implications. The University of Chicago Press.
- Faruqi, Mariam. 2011. "A Question of Faith: A Report on the Status of Religious Minorities in Pakistan." *Jinnah Institute*.
- Fraser, Erik. 2010. "The Dalits of India: education and development." *E-International Relations*, June 23. Retrieved on June 28, 2021. <a href="https://www.e-ir.info/2010/06/23/the-dalits-of-india-education-and-development/">https://www.e-ir.info/2010/06/23/the-dalits-of-india-education-and-development/</a>
- Gazdar, Haris and Hussain B. Mallah. 2012. "Class, Caste and Housing in Rural Pakistani Punjab: The Untold Story of the Five Marla Scheme." *Contributions to Indian Sociology* 46(3):311–36. https://doi.org/10.1177/006996671204600303
- Gazdar, Haris. 2007. "Class, Caste or Race: Veils over Social Oppression in Pakistan." *Economic and Political Weekly* 42(2):86-88. <a href="https://www.jstor.org/stable/4419123">https://www.jstor.org/stable/4419123</a>
- Geeta, Kingdon. 2009. "The Quality and Efficiency of Private and Public Education: A Case-Study of Urban India." Oxford Bulletin of Economics and Statistics. <a href="https://doi.org/10.1111/j.1468-0084.1996.mp58001004.x">https://doi.org/10.1111/j.1468-0084.1996.mp58001004.x</a>
- Ghani, Faras. 2017. "Veeru Kohli: A bonded laborer who took on the Bhuttos." *Al Jazeera*, April 13. Retrieved on June 28, 2021.

- https://www.aljazeera.com/features/2017/4/13/veeru-kohli-a-bonded-labourer-who-took-on-the-bhuttos
- Groot, Afke D. 2007. "Deprived Children and Education." IREWOC.
- Hasan, Arif. 2009. The Unplanned Revolution: Observations on the Processes of Socio-economic Change in Pakistan. (2nd Edition ed.). Karachi: Oxford University Press.
- Hossain, Mohammad S. 2016. "Multiple Deprivations: Schooling Experience of Dalit Children in Bangladesh." *International Institutes of Social Studies*. Retrieved on June 20, 2021. <a href="https://thesis.eur.nl/pub/37340/MHossain\_MA\_2015\_16\_SJP.pdf">https://thesis.eur.nl/pub/37340/MHossain\_MA\_2015\_16\_SJP.pdf</a>
- Husain, Zakir. 2005. "Analysing demand for primary education: Slum dwellers of Kolkata." *Economic and Political Weekly* XL(2):137-147.
- Hussain, Ghulam. 2014. *The Rise of Ethnic Peasant in Sindh*. Qauid-i-Azam Univeristy Islambad. Retrieved from <a href="https://www.academia.edu/13684482/The\_Rise\_of\_Ethnic\_Peasant\_in\_Sindh\_Book\_Title">https://www.academia.edu/13684482/The\_Rise\_of\_Ethnic\_Peasant\_in\_Sindh\_Book\_Title</a>
- International Labor Organization. 2005. "Combating Child Labor in Carpet Industry (Phase II)." International Programme on Elimination of Child Labor, Islamabad, Pakistan. Retrieved on August 10, 2021.

- http://www.ilo.org/public/english/region/asro/newdelhi/ipec/responses/pakistan/p2.htm.
- Islam, Monjurul., Afsana Amin, and Zakir Hossain. 2015. "Challenges and prospects for Dalits securing their right to education in Bangladesh." *Dhaka: Nagorik Uddog & Bangladesh Dalit and Excluded Rights Movement 2014.*
- Jain, Lakshmi C. 1981. "Emancipation of Scheduled Castes and Tribes: Some Suggestions." *Economic and Political Weekly* 16(9):325-332.
- Kabeer, Naila. 2006. "Poverty, Social Exclusion and the MDGs: The Challenge of 'Durable Inequalities' in the Asian Context." *IDS Bulletin* 37(3):64-78. <a href="https://doi.org/10.1111/j.1759-5436.2006.tb00270.x">https://doi.org/10.1111/j.1759-5436.2006.tb00270.x</a>
- Kakde, Suresh. 2008. "Globalization and Scheduled caste, in KaradeJagan (ed.), Development of Scheduled Castes and Scheduled Tribes in India." *Cambridge Scholars Publishing, U.K.* 10-18. Retrieved on June 23, 2021, from <a href="https://www.academia.edu/6163636/Development of Scheduled">https://www.academia.edu/6163636/Development of Scheduled</a> Castes and Scheduled Tribes in India
- Khan, Akhtar H. and Naushin Mahmood. 1997. "Education in Pakistan: Fifty Years of Neglect." *Pakistan Institute of Development*

- Economics, Islamabad. 36(4):647-667. https://www.jstor.org/stable/41260063
- Kirpal, Viney. 1978. "Higher Education for the Scheduled Castes and Scheduled Tribes." *Economic and Political Weekly*. <a href="https://www.jstor.org/stable/4366342">https://www.jstor.org/stable/4366342</a>
- Kolhi, Chander. 2014. Drought and migration: A brief story of Tharparkar.

  \*Daily Times\*, March 23. Retrieved on September 12, 2021.

  https://dailytimes.com.pk/105392/drought-and-migration-a-briefstory-of-tharparkar/
- Kulkarni, Suresh D. 1974. "Scheduled Castes and Scheduled Tribes—
  Problems and Policies." *Economic and Political Weekly*.

  <a href="https://www.jstor.org/stable/4364095">https://www.jstor.org/stable/4364095</a>
- Mabeya, Danvas O. 2018. Employment: A significant indicator of socioeconomic upward mobility for the lost boys living in the Kansas
  City area. Immigration and Development, InTech.
  https://books.google.com/books?hl=en&lr=&id=7GQDwAAQBAJ&oi=fnd&pg=PA89&ots=kA7FGzlfvs&sig=CQAf
  mAeGgkW1t6rdccuq3CyL3AY
- Malik, Abdullah. 2018. "Activist Veerji Kohli talks about the challenges he has faced for raising voice against tyranny." *Daily Times*, March 23. Retrieved on June 30, 2021.

- https://dailytimes.com.pk/218398/activist-veerji-kohli-talks-about-the-challenges-he-has-faced-for-raising-voice-againt-tyranny/
- Malik, Dr. Iftikhar H. 2002. *Religious Minorities in Pakistan*. Pakistan's National Council for Justice and Peace (NCJP), Minority Rights Group International, n.d.
- Malik, Suniela. 1979. "Social Integration of Scheduled Castes." *Abhinav Publications, New Delhi*.
- Mehdi, Ajmal M. 2019. "Tharparkar the incomparable backward region of Pakistan." *Daily Times*, April 10. Retrieved on June 19, 2021. https://dailytimes.com.pk/374856/tharparkar-the-incomparable-backward-region-of-pakistan
- Memon, Ghulam R. 2007. "Education in Pakistan: The Key Issues, Problems, and The New Challenges." *Journal of Management and Social Sciences* 3(1):47-55.

  <a href="http://www.ilmauniversity.edu.pk/assets/custom/journal/jbs/3.1/5">http://www.ilmauniversity.edu.pk/assets/custom/journal/jbs/3.1/5</a>.

  OChallenges.pdf</a>
- Naik, Raqib H. 2021. "California student body demands ban on castebased discrimination." *Al Jazeera*, April 16. Retrieved on June 28, 2021. <a href="https://www.aljazeera.com/news/2021/4/16/california-student-body-demands-ban-on-caste-based-discrimination">https://www.aljazeera.com/news/2021/4/16/california-student-body-demands-ban-on-caste-based-discrimination</a>

- Nambissan, Geetha . 1996. "Equity in Education? Schooling of Dalit

  Children in India." *Economic and Political Weekly*31(16/17):1011-1024. https://www.jstor.org/stable/4404063
- Naz, Arab., Hazir Ullah., Qaisar Khan., Waseem Khan, and Umar Daraz.

  2014. "A Critical Discourse of Child Victimization and Abuses through Labor in Pakistan." *Pakistan Journal of Criminology* 6(1):123-141.
- Pakistan Dalit Solidarity Network (PDSN) and International Dalit Solidarity Network (IDSN). *Scheduled Caste Women in Pakistan*.

  Alternative report to the UN Committee on the Elimination of all Forms of Discrimination Against Women (CEDAW) for the examination of the 4th periodic report of Pakistan at the 54<sup>th</sup> CEDAW session in February 2013.
- Parul. 2014. "Disparity in Higher Education: The Context of Scheduled

  Castes in Indian Society." International Journal of Research in

  Humanities, Arts and Literature.

  <a href="https://www.impactjournals.us/index.php/download/archives/2-11-1395844860-5.%20Humanities-">https://www.impactjournals.us/index.php/download/archives/2-11-1395844860-5.%20Humanities-</a>

  Disparity%20in%20Higher%20Education%20The%20Context-Parul.pdf
- Patel, Shaista A. A. 2020. "It is time to talk about caste in Pakistan and Pakistani diaspora." *Al Jazeera*, December 15. Retrieved on June

- 28, 2021. <a href="https://www.aljazeera.com/opinions/2020/12/15/it-is-time-to-talk-about-caste-in-pakistan-and-pakistani-diaspora">https://www.aljazeera.com/opinions/2020/12/15/it-is-time-to-talk-about-caste-in-pakistan-and-pakistani-diaspora</a>
- Perrigo, Billy. 2020. "As India's Constitution Turns 70, Opposing Sides

  Fight to Claim Its Author as One of Their Own." *Time*, January 24.

  Retrieved on September 12, 2021. <a href="https://time.com/5770511/india-protests-br-ambedkar/">https://time.com/5770511/india-protests-br-ambedkar/</a>
- Qadeer, Mohammad. 2012. "Pakistan social and cultural transformations in a Muslim Nation." *Routledge*.
- Ravi, Babu M. and T. Chandrasekarayy. 2015. "Education status and its impact on the development of scheduled castes: an overview."

  International Journal of Multidisciplinary Research and Development.

http://www.allsubjectjournal.com/vol2/issue1/PartG/pdf/44.1.pdf

- Rehman, Ibn A. 2015. "The miserable scheduled castes." *The Dawn*, June 18. Retrieved on June 19, 2021. https://www.dawn.com/news/1188782/
- Rehman, Zia U. 2017. "Scheduled castes have a separate box for them, but only if anybody knew." *The News*, April 10. Retrieved on June 21, 2021. <a href="https://www.thenews.com.pk/print/197657-Scheduled-castes-have-a-separate-box-for-them-but-only-if-anybody-knew">https://www.thenews.com.pk/print/197657-Scheduled-castes-have-a-separate-box-for-them-but-only-if-anybody-knew</a>

- Samuel, Persis and Muhammad S. Nisar. 2020. "Inequalities and opportunity gaps for two religious' groups." *Elsevier*. Retrieved on June 20, 2021. from <a href="https://doi.org/10.1016/j.cities.2020.103098">https://doi.org/10.1016/j.cities.2020.103098</a>
- Sen, Amartya. 2000. "Social Exclusion: Concept, Application, and Scrutiny." *Asian Development Bank, Manila*. http://refhub.elsevier.com/S0738-0593(15)00049-8/sbref0220
- Shah, Dawood. 2003. "Decentralization in the Education System of Pakistan: Policies and Strategies". Academy of Educational Planning and Management Ministry of Education, Government of Pakistan, Islamabad. Retrieved on July 6, 2021. <a href="https://www.academia.edu/3665909/Decentralization">https://www.academia.edu/3665909/Decentralization</a> in the Education System of Pakistan
- Shah, Zulfiqar. 2007. Long Behind Schedule: A Study on Plight of Scheduled Caste Hindus in Pakistan. Indian Institute of Dalit Studies (IIDS) & International Dalit Solidarity Network (IDSN).

  TRDP & PILER, December.
- Stromquist, Nelly P. and Gustavo E. Fischman. 2009. "Introduction From Denouncing Gender Inequities to Undoing Gender in Education:

  Practices and Programmes Toward Change in The Social Relations of Gender." International Review of Education / Internationale

  Zeitschrift Für Erziehungswissenschaft / Revue Internationale del'

  Education 55(5/6):463–82. http://www.jstor.org/stable/40608073

- Talik, Jandhyala . G. 1979. "Inequality in Education in India." *Indian Journal of Industrial Relations* 14(3). Retrieved on June 23, 2021.
- Tamim, Tayyaba and Hana Tariq. 2015. "The intersection of caste, social exclusion, and educational opportunity in rural Punjab."

  International Journal of Educational Development 43:51–62.

  <a href="http://dx.doi.org/10.1016/j.ijedudev.2015.04.008">http://dx.doi.org/10.1016/j.ijedudev.2015.04.008</a>
- The News. 2012. "Call to protect scheduled castes' constitutional rights."

  Retrieved on June 19, 2021.

  <a href="https://www.thenews.com.pk/archive/print/380339-call-to-protect-scheduled-castes-constitutional-rights/">https://www.thenews.com.pk/archive/print/380339-call-to-protect-scheduled-castes-constitutional-rights/</a>
- Ullah, Hazir and Johar Ali. 2018. "Schools and Families: Reproduction of Class Hierarchies through Education in Pakistan." *Pakistan Journal of Criminology* 10(03):91-111.
- UNICEF and UNESCO. 2007. A Human Rights-Based Approach to Education. New York: United Nations Children's Fund, and Paris:

  United Nations Educational, Scientific and Cultural Organization.

  <a href="https://www.right-to-education.org/sites/right-to-education.org/files/resource-attachments/A%20Human%20Rights-based%20Approach%20to%20Education%20for%20All\_0.pdf">https://www.right-to-education.org/sites/right-to-education.org/files/resource-attachments/A%20Human%20Rights-based%20Approach%20to%20Education%20for%20All\_0.pdf</a>
- USAID. 2008. Education from a Gender Equality Perspective.
- Vasavi, Aninhalli R. 2012. "Government rahmin: Caste, the educated unemployed, and the reproduction of inequalities." *TRG Poverty &*

Education I Working Paper Series, Max Weber Stiftung,
Foundation German Humanities Abroad. Retrieved from https://www.ghil.ac.uk/fileadmin/redaktion/dokumente/trg\_india/P
aper 1 A R Vasavi.pdf

#### **ANNEXURE**

# The Education of Kolhi Community in Sindh: Opportunities and Challenges



#### Aakash

# Department of Sociology, Quaid-i-Azam University, Islamabad, Pakistan

I am **Aakash**, a research student in the Department of Sociology, Quaid-i-Azam University, Islamabad, I am researching on the topic, "**The Education of the Kolhi Community: Opportunities and Challenges.** Your responses will be kept confidential and will be used only for academic purposes.

# **Background Information**

1.	Age of the respondent
2.	Education level of the respondent
3.	Occupation/Profession of the respondent
4.	Respondents, Ethnicity

Questions	Agree	Strongly Agree	Neutral	Disagree	Strongly Disagree
1. The school is quite away from your home.					
2. Your parents want you to get education.					
3. Your family has enough money to support your education.					
4. You work as labor after school time.					
5. Your family has enough money to buy books and copies.					
6. Your parents have financial support to send you for secondary education.					
7. Your parents want that you should work with them on the land.					
8. Your parents expect that you will get a job after completing matric/intermediate and there is no need for further education.					
9. Due to a lack of job opportunities, you prefer to work on agricultural land.					
10. Your family migrates from one place to another for seasonal earning.					
11. You cannot attend the school continuously, due to seasonal migration.					
12. Do you think that your parents do your marriage during your studies at the school level?					
13. Your teachers discriminate you on the basis of your religion.					
14. Your teachers do not respond to you then they respond to Muslim fellows.					
15. You feel discrimination from your Muslim fellows in class.					
16. Your glass and plates are separate at the school canteen.					
17. You are allowed to eat at the same food points in school.					